Author's response to reviews

Title: Tongue lesions in psoriasis: A controlled study.

Authors:

Maryam Daneshpazhooh (maryamdanesh@yahoo.com)
Homayoon Moslehi (moslehi_h@yahoo.com)
Maryam Akhyani (ma_akhyani@yahoo.com)
Marjan Etesami (marjanetesami@yahoo.com)

Version: 2 Date: 30 August 2004

Author's response to reviews: see over
30 August, 2004

Dear Editors,

In response to your letter dated 9 August, 2004, I would like to submit the revised version of our manuscript (MS#1519039183401467) entitled “Tongue lesions in psoriasis: A controlled study”. In accordance with the referee’s comment, we have changed the conclusions of the abstract, as well as the conclusions of our manuscript. I hope the revised version of our manuscript will be judged ready for publication and look forward to hearing from you.

Sincerely yours,

Maryam Daneshpazhooh